Author’s response to reviews

Title: Invasive pneumococcal infections among persons with and without underlying medical conditions: implications for prevention strategies

Authors:

   Peter Klemets (peter.klemets@ktl.fi)
   Outi Lyytikäinen (outi.lyytikainen@ktl.fi)
   Petri Ruutu (petri.ruutu@ktl.fi)
   Jukka Ollgren (jukka.ollgren@ktl.fi)
   J Pekka Nuorti (PNuorti@cdc.gov)

Version: 8 Date: 18 July 2008

Author’s response to reviews: see over
The Editor,
BMC Infectious Diseases
Melissa Norton, MD
BioMed Central Ltd,
Middlesex House,
34-42 Cleveland Street,
London W1T 4LB, UK.

July 18, 2008

Dear Dr. Norton,

Enclosed is a revised version of our manuscript titled “Invasive pneumococcal infections among persons with and without underlying medical conditions: implications for prevention strategies” (MS: 2062356805195363), which we would like to resubmit for publication in the BMC Infectious Diseases in the category of research article. We have made some minor changes to the format of some tables of the manuscript.

Thank you for considering this paper for the BMC Infectious Diseases. Please address correspondence regarding this manuscript at the address below.

Sincerely yours,

Peter Klemets, MD
Department of Infectious Disease Epidemiology and Control
National Public Health Institute,
Mannerheimintie 166
FIN-00300 Helsinki, Finland
Phone: 358-9-47448778
Fax: 358-9-47448468
E-mail: peter.klemets@ktl.fi

J. Pekka Nuorti, MD
National Center for Immunization and Respiratory Diseases,
1600 Clifton Road, NE, Mailstop C-23
Centers for Disease Control and Prevention (CDC),
Atlanta, GA 30333, U.S.A.
Email: PNuorti@cdc.gov
A response to the comment by reviewer 3

Reviewer 3: Åke Örtqvist

Reviewer's report:

The manuscript is now fully acceptable. I have only one question referring the last reference (No. 45), which does not concern the situation in Europe, but the etiology of CAP in Chile, and does not refer to the effects of pneumococcal conjugate vaccination - this must be a mistake?

RESPONSE:

Reference #45 was erroneous due to a mistake. It is now replaced by the correct reference: